Author’s response to reviews

Title: Should heart age calculators be used alongside absolute cardiovascular disease risk assessment?

Authors:
Carissa Bonner (carissa.bonner@sydney.edu.au)
Katy Bell (katy.bell@sydney.edu.au)
Jesse Jansen (jesse.jansen@sydney.edu.au)
Paul Glasziou (pglaszio@bond.edu.au)
Les Irwig (les.irwig@sydney.edu.au)
Jenny Doust (jdoust@bond.edu.au)
Kirsten McCaffery (kirsten.mccaffery@sydney.edu.au)

Version: 2 Date: 15 Jan 2018

Author’s response to reviews:

Dear Dr Dickens,

Thank you for considering our revised manuscript BCAR-D-17-00569: Should heart age calculators be used alongside absolute cardiovascular disease risk assessment?

We have responded to the remaining issues as follows:

1. Please confirm whether you received permission to use figures 1 and 2?

These figures were created by the authors using edited screen shots of publicly available websites, so they are not subject to copyright permissions.

2. Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.

This section has been added on page 21.
3. Pleased remove the version of the manuscript with marked changes.

This version has been removed.

4. Please ensure that the ‘Author contributions’ section only contains information regarding the role of each author. Please move any additional author information to its own section ‘Author information’. This section can be added to the end of the declarations.

This change has been made on page 22.

5. In the Funding section, please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

This information has been added on page 22.

6. Please ensure that your manuscript is formatted as per the submission guidelines. For example the abstract should include a background, main body of the abstract and short conclusion.

The manuscript has been checked and edited to ensure it meets the submission guidelines, including correct subheadings and references in square brackets.

We hope the revised manuscript is now suitable for publication in your journal.

Sincerely,

Dr Carissa Bonner, on behalf of all authors